Author's response to reviews

Title: Lack of association between the chemokine receptor 5 polymorphism CCR5delta32 in rheumatoid arthritis and juvenile idiopathic arthritis

Authors:

Ewald Lindner (Ewald.Lindner@student.uibk.ac.at)
Gry BN Nordang (g.b.namlos@medisin.uio.no)
Espen Melum (espen.melum@medisin.uio.no)
Berit Flato (berit.flato@rikshospitalet.no)
Anne Marit Selvaag (aselvaag@ulrik.uio.no)
Erik Thorsby (erik.thorsby@medisin.uio.no)
Tore K Kvien (t.k.kvien@medisin.uio.no)
Oystein T Forre (o.t.forre@medisin.uio.no)
Benedicte A Lie (b.a.lie@medisin.uio.no)

Version: 4 Date: 7 June 2007

Author's response to reviews:

Dear Editor,

We have made the final formatting changes as requested by email.

The website - see page 5 - have been removed from the text and and is now cited from the main reference list. We have used the reference software Reference Manager 11.

I'm sorry this was not corrected in the last revised manuscript.

Sincerely yours,

Gry B. N. Nordang